Author's response to reviews

Title: Low prevalence of 'classical' microscopic colitis but evidence of microscopic inflammation in Asian patients with diarrhoea predominant irritable bowel syndrome

Authors:

Ida Hilmi Dr (ida.hilmi@gmail.com)
Juanda Hartono Dr (juanda@gmail.com)
Jayalakshmi Pailoor Professor (jpailoor@um.edu.my)
Sanjiv Mahadeva Professor (sanjiv@ummc.edu.my)
Khean-Lee Goh Professor (gohkl@um.edu.my)

Version: 2 Date: 11 September 2012

Author's response to reviews:

Dear editor,

The study was approved by the local ethics committee (University of Malaya Medical Centre)

Informed consent was obtained in all patients.

The changes for the figure legends have been done.

Regards,

Ida Hilmi